Author’s response to reviews

Title: Coracoid Abnormalities and Their Relationship with Glenohumeral Deformities in Children with Obstetric Brachial Plexus Injury

Authors:

Rahul K Nath (drnath@drnathmedical.com)
Faiz Mahmooduddin (faiz@drnathmedical.com)
Xiaomei Liu (xiaomei@drnathmedical.com)
Melissa J Wentz (melissa@drnathmedical.com)
Andrea D Humphries (andrea@drnathmedical.com)

Version: 3 Date: 5 October 2010

Author’s response to reviews: see over
October 5, 2010

Melissa Norton, M.D.
Editor-in-Chief
BMC Musculoskeletal Disorders

Dear Dr. Norton,

Thank you for accepting our manuscript for publication in the *BMC Musculoskeletal Disorders*. Enclosed please find our revised manuscript entitled “Coracoid Abnormalities and Their Relationship with Glenohumeral Deformities in Children with Obstetric Brachial Plexus Injury”. As instructed, here we provide our responses (in red) to the editor’s concerns, point-by-point:

Editorial comments:

Formatting changes to be made:

Major revisions (we require the author to make these changes)

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods, Results, Conclusions

Done

Figures: Please provide figure titles. These should be concise and be the first sentence of the figure legend.

Done

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Removed

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: [http://www.biomedcentral.com/info/ifora/figures](http://www.biomedcentral.com/info/ifora/figures)

Done

Figures: Please ensure all mention of figures is consistent (it should be Figure or Fig but not a mixture of both)
Corrected

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Done

Sincerely,

Rahul K. Nath, M.D.
Texas Nerve and Paralysis Institute
6400 Fannin Street, Suite 2420
Houston, TX 77030 USA
Email: nath@drnathmedical.com